Author's response to reviews

Title: Factors Affecting Medication Adherence: Patient Perspectives from Five Veterans Affairs Facilities

Authors:

Clarissa Hsu (hsu.c@ghc.org)
Jaclyn M Lemon (lemonj@u.washington.edu)
Edwin S Wong (edwin.wong@va.gov)
Elizabeth Carson-Cheng (eli_carson@yahoo.com)
Mark Perkins (mark.perkins2@va.gov)
Margaret S. Nordstrom (margaretnordstrom@gmail.com)
Chuan-Fen Liu (Chuan-Fen.Liu@va.gov)
Carol Sprague (carol.sprague@va.gov)
Christopher L. Bryson (Christopher.Bryson@va.gov)

Version: 2 Date: 21 January 2014

Author's response to reviews:

January 21, 2014

Flory Mae Calumpita
Journal Editorial Office

Dear Editorial Office,

We are pleased to re-submit our manuscript entitled “Factors Affecting Medication Adherence: Patient Perspectives from Five Veterans Affairs Facilities,” for consideration as an original article in BMC Health Services Research.

Upon receipt of the initial review from the BMC Health Services Research editorial staff the following changes and modifications to the submission have been made.

- The corresponding author has been changed to Clarissa Hsu.
- The abstract was updated and the “Research Design and Methods” header was removed and replaced with the header of “Methods.”
- The background is 249 words in length.
- The research article headings were updated by replacing “Research Design and Methods” with the required heading of “Methods.”
- A list of abbreviations was added with the inclusion of two abbreviations. The study was approved by the VA Puget Sound Health Care System Research & Development Committee as well as the Institutional Review Board.
We look forward to your further review and decision.

Sincerely,

Clarissa Hsu, PhD
Research Associate/ Research Program Manager
Center for Community Health and Evaluation, Group Health Research Institute
1730 Minor Ave, Suite 1600
Seattle, WA 98103
Phone 206-287-4276
E-mail hsu.c@ghc.org
www.cche.org